Author's response to reviews

Title: Intraparenchymal meningioma mimicking cavernous malformation: case report and literature review

Authors:

Senol sd Jadik MD (Senol.Jadik@uksh.de)  
Alexandru Ac C. Stan MD (alex.stan@gmx.net)  
Uwe ud Dietrich MD (dietrich@radiologie.gilead.de)  
Terttu ta A. Pietilä Prof. (Terttu.Pietilae@evkb.de)  
Alaa Eldin AE Elsharkawy MD (alaa.elsharkawy@googlemail.com)

Version: 5  
Date: 15 November 2014

Author's response to reviews:

Dear Kathrina,

We hereby our manuscript revision “Intraparenchymal meningioma mimicking cavernous malformation: case report and literature review”

All changes have been highlighted in the revised manuscript.

We hope that our revision satisfied all reviewers. Sincerely yours

Dr. MD. Alaa Elsharkawy

13-11-2014

Reviewer #1:

Comment: 1

• This is a case report of intraparenchymal meningioma which might pose diagnostic challenge. Essential information was contained in the manuscript but I’d appreciate if the CT image can also be included for completion sake.

Authors comment

• Adding a new image will make the case report very long, in addition the needed information already exists in the MRI images.

Comment: 2

• Literature review, P.5, Line 9 “……insufficient data und cases……”

Authors comment

• We have improved this paragraph

Comment: 3

• Discussion, P.6, Line 9 “…… and within the region within the Sylvian fissureare……”
Authors comment
• We are sorry for this mistake we corrected it

Comment: 4
• Figure legend: Figure 6 “…… horl structures……”

Authors comment
• We are sorry for this mistake we corrected it.

Comment: 5
• Table 1: “Sold mass”? solid mass

Authors comment
• We are sorry for this mistake we corrected it

Comment: 5
• Case presentation, P.3, Line 2 “…… no neurological deficits were present.”

Authors comment
• We are sorry for this mistake we corrected it

Comment: 6
• Case presentation, P.4, Line 18 “……(at low power (5x objectives)__. A close bracket is needed.

Authors comment
A close bracket has been added.

We hope that our revision become more satisfactory for you

13-11 -2014 sincerely yours
Alaa Elsharkawy